Author’s response to reviews

Title: Neuromyelitis optica spectrum disorders without and with autoimmune diseases

Authors:

Bingjun Zhang (zhangbj@outlook.com)
Yi Zhong (6372636@qq.com)
Yanqiang Wang (wangqiangdoctor@126.com)
Yongqiang Dai (dyq_ny@163.com)
Wei Qiu (qw9406@gmail.com)
Lei Zhang (13726209240@139.com)
Haiyan Li (190494812@qq.com)
Zhengqi Lu (lzqzssy@gmail.com)

Version: 9 Date: 30 July 2014

Author’s response to reviews: see over
Response

MS: 1045091991290486
Neuromyelitis optica spectrum disorders without and with autoimmune diseases
Bingjun Zhang, Yi Zhong, Yanqiang Wang, Yongqiang Dai, Wei Qiu, Lei Zhang, Haiyan Li and Zhengqi Lu

Dear Dr. Handel and Santos,

Thank you very much for your letter and advice. We have revised the manuscript, and would like to re-submit it for your consideration. Point by point responses to your comments are listed below this letter.

We hope that the revised version of the manuscript is now acceptable for publication in your journal.

I look forward to hearing from you soon.

With best wishes,

Zhengqi Lu
Department of Neurology
The Third Affiliated Hospital, Sun Yat-sen University,
600 Tianhe Road, Guangzhou, Guangdong 510630 China.
Tel/Fax: +86-20-8751-6822
Email: lzqzssy@gmail.com
We would like to express our sincere thanks to the editors and reviewers for the constructive and positive comments.

Please include the email addresses of all the authors on the title page.

Answer: The email addresses have been added in the title page of the revised version to address this issue.